Author's response to reviews

Title: Adoption of computerized provider order entry systems: An organization-wide study based on diffusion of innovations theory

Authors:

Bahlool Rahimi (bahra@ida.liu.se)
Vivian Vimarlund (vivvi@ida.liu.se)
Toomas Timpka (tti@ida.liu.se)
Srinivas Uppugunduri (Srinivas.Uppugunduri@lio.se)
Mikael Svensson (Mikael.Svensson@lio.se)

Version: 2 Date: 7 July 2009

Author's response to reviews:

BMC Medical informatics and decision making journal scope is to publish articles on relation to the implementation, use, and evaluation of health information technologies and decision-making within the healthcare setting. Because my article is about implementation and use of CPOE systems and we have analyzed physicians and nurses attitudes regarding adoption of CPOE system and since your journal are open access and my intention is that my article is accessed to every body to be used and cited, I believe that my article is a relevant article to be considered for publication in BMC Medical informatics and decision making journal.